LEVEL OF PHYSICAL ACTIVITY IS POSITIVELY CORRELATED WITH PERCEIVED IMPACT ON LIFE 12 MONTHS AFTER STROKE: A CROSS-SECTIONAL STUDY

Peta COOK, RPT, Katharina S. SUNNERHAGEN, MD, PhD and Hanna C. PERSSON, RPT, PhD From the Department of Clinical Neuroscience, Institute of Neuroscience and Physiology, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden

Objectives: To examine the relationship between, and impact of, level of physical activity and perceived impact on life at 12 months post-stroke. **Design:** Cross-sectional study.

Subjects: A total of 73 participants with first-time stroke included in the Stroke Arm Longitudinal study at the University of Gothenburg (SALGOT study), Sweden.

Methods: Perceived impact of stroke was assessed with the Stroke Impact Scale and level of physical activity was assessed with the Saltin-Grimby Physical Activity Scale at 12 months post-stroke. Data were presented with descriptive and logistic regression analyses.

Results: The physically active group perceived their strength, emotion, mobility, participation and overall stroke recovery as significantly less problematic compared with the inactive group. Being physically active contributed to higher scores in the Strength domain (odds ratio, OR 7.89) and in the Stroke Recovery domain (OR 18.55). In the Participation domain being physically active (OR 8.01) and independent (OR 0.162) contributed to higher scores.

Conclusion: A positive correlation was found between level of physical activity at 12 months poststroke and levels of strength, participation and stroke recovery.

Key words: cerebrovascular accident; exercise; rehabilitation; self-concept; recovery of function; Stroke Impact Scale.

Accepted Mar 3, 2020; Epub ahead of print Mar 16, 2020

J Rehabil Med 2020; 52: jrm00056

Correspondence address: Peta Cook, Research Group of Rehabilitation Medicine, Per Dubbsgatan 14, Floor 3, 41345 Gothenburg, Sweden. E-mail: peta.cook@vgregion.se

Physical activity (PA) has many health benefits, such as reducing the risk of heart disease, cancer and stroke (1). An adequate level of PA for adults is defined as a minimum of 150 min/week of moderate intensity aerobic activity together with regular strength training (2), either through structured activities or incidental PA. However, despite worldwide promotion of PA (1), not only does inadequate PA remain one of the 5 most preventable risk factors of stroke, PA levels post-stroke are lower than in the comparable healthy population (3).

LAY ABSTRACT

It is common for stroke survivors to experience longterm symptoms. This study investigated individuals' perceptions of how stroke symptoms impact on their lives, and examined the connection between this and their level of physical activity 12 months after stroke. A total of 73 stroke survivors completed 2 questionnaires; about their level of physical activity, and their perception of how their life is impacted by stroke. The results showed that a higher level of physical activity at 12 months after stroke was linked to stroke survivors perceiving that their strength, participation in life and overall recovery from stroke was better. This study highlights the importance of being physically active even long-term after stroke, as this can help to optimize the perception of recovery and reduce the impact of stroke on life.

Stroke symptoms are many and varied; for example, reduced motor function, coordination, sensation and cognitive function (4). Strong evidence has been found for the use of task-specific training within stroke rehabilitation in order to improve functions such as muscle strength, walking ability and aerobic fitness (5–8). Research has also found that higher levels of PA alone can improve functional skills, such as walking ability, balance and physical fitness post-stroke (3). However, conversely, research has shown that physical function alone cannot explain the level of PA post-stroke (9–11).

Examining outcomes post-stroke is complex and, as discussed by the International Classification of Function, Disability and Health (ICF), examining multiple dimensions of health is critical (12). Multidimensional measurement tools, such as the stroke-specific Stroke Impact Scale (SIS) (13, 14), which examines varied outcomes such as function, activity levels and participation levels, and are essential in order to examine recovery post-stroke. Furthermore, it is well established that individual perception of health is important (15), which supports the use of self-rater scales such as SIS.

Increased PA has many physical and psychological benefits post-stroke (3, 8, 16, 17). However, energy expenditure has been shown to be higher in the stroke population (18), which in turn could support a hypothesis that higher levels of PA lead to increased fatigue and the consequential perceived impact of stroke on life is greater. It is thus important to develop an im-

p. 2 of 6 P. Cook et al.

proved understanding of the relationship between level of PA and self-perceived impact of stroke on life, as it can have a significant impact on long-term stroke rehabilitation.

The aim of this study is to examine the relationship between, and impact of, level of PA and perceived impact on life at 12 months post-stroke.

METHODS

During an 18-month period (2009–2010) individuals with firsttime stroke who were cared for at the stroke unit of Sahlgrenska University Hospital were included consecutively into the Stroke Arm Longitudinal Study at the University of Gothenburg (the SALGOT study) (19).

Inclusion from the SALGOT study (19) were patients who were: diagnosed with a first-time stroke; had reduced arm and hand function on day 3 post-stroke according to the Action Reach Arm Test (ARAT), with a maximum score of 57; were admitted to the stroke unit within 3 days of onset of stroke; were living in the Gothenburg urban area (maximum 35 km from Sahlgrenska University Hospital); and were over 18 years of age. The exclusion criteria were: previously reduced function in the upper limb prior to stroke (due to injury or other condition); and non-Swedish speaking prior to stroke (19). Further inclusion criteria for the current study included: completion of both Saltin-Grimby Physical Activity Level 6-Scale (SGPALS) (20, 21) and SIS at assessment 12 months post-stroke. This resulted in a total of 73 participants (see Fig. 1).

The perceived impact of stroke was assessed with the selfassessment questionnaire SIS 3.0 (14). The SIS is used to assess the individual's perceived impact of stroke in 59 items, across 8 different domains (strength, memory, emotions, communication, activities of daily living, mobility, hand function and participation). Each item was scored in an ordinal scale (1–5), and summarized to a domain score. Each domain score was transformed using the following equation: domain score= (mean item score–1)/5–1 × 100, with each domain receiving



Fig. 1. Flow chart of the patient selection procedure, starting from the Stroke Arm Longitudinal Study at the University of Gothenburg (SALGOT study) population, and thereafter describing the flow of inclusion in the current study, at 12 months post-stroke. SIS: Stroke Impact Scale; SGPALS: Saltin-Grimby Physical Activity Level Scale.

a final total score out of 100 (14). SIS also includes one final question regarding overall stroke recovery (SR), scored 0-100 in a visual analogue scale (VAS). In the present study domains 1 (strength), 3 (emotion), 6 (mobility), 8 (participation), and SR were selected for analysis, as these domains are expected to be closely related to PA.

At 12 months post-stroke participants measured their level of PA during the past 6 months using SGPALS, a self-assessed 6-grade ordinal scale, ranging from physically inactive to an elite level of PA and training (20, 21). Based on the level of PA at 12 months the study population was divided into 2 groups: an inactive group (equivalent to SGPALS 1–2) and an active group (equivalent to SGPALS 3–6) (21).

Sociodemographic data from the time of onset of stroke was used to describe the population.

Medical data from the acute stages was collected via the patients' medical record as well as being supplemented by Riksstroke (The Swedish Stroke Register) (19, 22). Medical data included type of stroke (ischaemic or haemorrhagic) (23), and level of dependency (24). Level of dependency in activities of daily living was measured by modified Rankin Scale (mRS) (25) at time of discharge from hospital.

Variables that are shown to be important for PA and/or stroke outcome were assessed in the study; including age (24), sex (26), smoking status ("yes" corresponds with ≥ 1 cigarette/day, or quit during the last 3 months) (27), previous cerebrovascular disease (CVD) ("yes" corresponds to diagnosis of at least one of the following diseases prior to stroke: heart failure, ischaemic heart failure, hypertension, angina, hypercholesterolaemia, aortic insufficiency, atrial fibrillation, myocardial infarct, prosthetic heart valve, or rheumatic valvular heart disease) (27), living arrangements ("living alone" or "sharing household with another adult") and occupational status ("working" \geq 75% fulltime work, or "not working" <75%) (28).

All assessments were completed by 3 experienced physiotherapists in a standardized procedure (SALGOT protocol) (19).

Data analysis

Differences between groups (active and inactive) in demographic data were analysed using Fisher's exact test and Mann–Whitney U test. The level of significance was set as p-value < 0.05.

Raw responses from SIS were initially dichotomized into 1–3 representing "problematic" symptoms, and 4–5 "not problematic" symptoms (14, 29). Based on this, the median score between 3 and 4 was calculated to be 62.5 and final dichotomization of the SR VAS score was dichotomized into 0–62 and 63–100.

Logistic regressions were used to examine the relationship between the different domains of SIS (dependent variable) and the level of PA (independent variable). The initial crosstabulation of each SIS domain with PA was conducted, and in the case of poor distribution (with fewer than 5 in a box) the variable was excluded from further analysis. Spearman's rankorder correlation (rho) was used to check all variables potentially included in the models, with rho set to rho ≥ 0.7 (30). If rho \geq 0.7, the variable was excluded from further analyses. In order to identify significant variables for further analysis, univariate regression of all remaining variables was completed with a significance level of p < 0.25, using Wald's test. Multivariate logistic regression was completed with the significant remaining variables (with p < 0.25). Previously excluded variables were then reinserted in the multivariate regression with level of significance set at p < 0.05 (Fig. 2). Resulting models were analyzed with: Omnibus test (significance < 0.5); Nagelkerke R square and Cox & Snell R-square test; as well as receiver operating

Table I. Clinical characteristic	s of the population	according to leve	l of physical activity, $n = 73$
----------------------------------	---------------------	-------------------	----------------------------------

	Active group	Inactive group		Total
Variable	n = 47	<i>n</i> = 26	<i>p</i> -value	n = 73
At onset				
Age years at onset			< 0.001	
Median (min-max)	64 (38-89)	77 (58-90)		67 (38–90)
Mean (SD)	62.7 (11.9)	74.3 (8.3)		66.8 (12)
Sex, M/F, n (%)	28/19 (59.6/40.4)	15/11 (57.7/42.3)	0.535	43/30 (58.9/41.1)
Ischaemic/haemorrhage, n (%)	36/11 (76.6/23.4)	24/2 (92.3/7.7)	0.083	60/13 (82.2/17.8)
Smoking, yes/no, n (%) $n = 72$	14/33 (29.8/70.2)	4/21 (26/84)	0.159	18/54 (25/75)
CVD: yes/no, n (%)	26/21 (55.3/44.7)	19/7 (73/27)	0.106	45/28 (61.6/38.4)
At discharge				
mRS 0/1/2/3/4/5, <i>n</i> (%)	0/0/9/21/17/0 (0/0/19.1/44.7/36.2)	0/0/0/7/19/0 (0/0/0/26.9/73.1)	0.001	0/0/9/28/36/0 (0/0/12.3/38.4/49.3)
At 12 months				
SGPALS 1/2/3/4/5/6, n (%)	0/0/21/23/2/1 (0/0/28.8/31.5/2.7/1.4)	12/14/0/0/0/0 (16.4/19.2/0/0/0/0)		12/14/21/23/2/1 (16.4/19.2/28.8/31.5/2.7/1.4)
Working/not working, n (%)	13/34 (27.7/72.3)	0/26 (0/100)	0.002	13/60 (17.8/82.2)
Living alone/living with someone else, n (%)	17/30 (36.2/63.8)	12/14 (46.2/53.8)	0.278	29/44 (39.7/60.3)

p < 0.05 was considered as a significant difference (in italic).

CVD: cardiovascular disease; mRS: Modified Rankin Scale; SGPALS: Saltin-Grimby Physical Activity Level Scale; SD: standard deviation.

characteristics (ROC) curves to assess for goodness of fit (30). Statistical analysis was completed using the IBM Statistical Package for Social Sciences (SPSS version 21.0, for Windows).

Ethical issues

Written information and consent was collected from all participants or from their closest relative. The SALGOT study applied for and was granted ethical approval by the Regional Ethics Testing Board in Gothenburg (5 May 2008, diary number 255-08) and the Declaration of Helsinki follows. The SALGOT study is registered on ClinicalTrials.gov (NCT01115348).

RESULTS

A total of 73 participants were included in the study (Fig. 1). The participants' mean age at stroke onset was 66.8 years (standard deviation (SD) 12), and 82.2% had an ischaemic stroke (Table I). Of the participants with ischaemic stroke, 9 received thrombolysis and one received thrombectomy. When comparing the active and inactive groups significant differences were present in age (p<0.001), level of dependency (mRS) (p=0.001) and working status (p=0.002) (Table I).

Table II. Number of participants corresponding to Stroke Impact

 Scale (SIS) domains divided into level of physical activity at 12 months

SIS domain		Active group, n=47,	Inactive group, n n=26, n	<i>p</i> -value
Strength (SIS-1)	Problematic	9	17	< 0.001
	Not problematic	38	9	
Emotion (SIS-3)	Problematic	4	9	0.008
	Not problematic	43	17	
Mobility (SIS-6)	Problematic	1	17	< 0.001
	Not problematic	46	9	
Participation (SIS-8)	Problematic	11	20	< 0.001
	Not problematic	36	6	
Stroke Recovery (SR)	Problematic	6	19	< 0.001
	Not problematic	41	7	

Participants in the active group perceived their strength, emotion, mobility, participation and SR as significantly less problematic compared with the in-active group (Table II).

Three of the 5 chosen SIS domains had sufficient data to build regression models; these were Strength (SIS-1), Participation (SIS-8) and SR (Fig. 2). No variables were found to correlate (rho ≥ 0.7) in any of the models, and all variables thus continued to the univariate regression models (Fig. 2). In the domain of Strength and SR, PA was the only independent variable that contributed to the final model, with odds ratio (OR) of 7.98 (SIS-1) and 18.55 (SR) (Table III). In the Participation domain the final model included PA (OR 8.01) and level of dependency (OR 0.162) (Table III). Nagelkerke R square and Cox & Snell R-square indicated that 19.3–26.5% of variation in the outcome SIS-1, 24.3–32.6% of the variation in outcome SIS-8, and 31.5–43.6% of the variability in SR was explained by their corresponding models.

DISCUSSION

This study examined the relationship between a PA and self-perceived impact of stroke, as measured by SIS, at 12 months post-stroke. The study found that PA has a significant and positive correlation with perceived strength (SIS-1) and overall stroke recovery (SR) at

Table III. Logistic regression modelling improvement in Strength,
Participation and Stroke Recovery

Dependent	Independent	OR	95% CI	<i>p</i> -value
Strength (SIS-1)	PA	7.98	2.69-23.64	0.001
Participation (SIS-8)	PA	8.01	2.35-27.244	0.001
	mRS	0.162	0.003-0.162	0.003
Stroke recovery (SR)	PA	18.55	5.48-62.73	0.001

Area under the curve according to ROC in each model: Strength 0.786, Participation 0.768, SR 0.802 CI: confidence interval; mRS: modified Rankin Scale; PA: physical activity;

CI: confidence interval; mRS: modified Rankin Scale; PA: physical activity; ROC: receiver operating characteristic.

JRM



Fig. 2. Flow chart of the analysis procedure for each Stroke Impact Scale (SIS) domain: starting at cross-tabulation, followed by each step of logistic regression analysis, and finally displaying the resulting models for each domain. CVD: cardiovascular diseases; PA: physical activity.

12 months post-stroke. Furthermore, PA combined with level of dependency has a significant and positive correlation with perceived participation (SIS-8) at 12 months post-stroke. That is, participants with a higher level of PA reported a lower perceived impact of stroke on their life at 12 months post-stroke, according to these 3 SIS domains.

The results of the current study imply that assisting stroke survivors to be physically active can have a positive impact on their life, as it is correlated with an increased perception of participation long term poststroke. A recent study (31) established that SIS-8 (participation) is sensitive to change over time, finding that almost 50% of their study population had a meaningful positive or negative change in this domain between 3 and 12 months post-stroke. This strengthens the reliability of SIS-8 and thereby supports our findings.

Guidetti et al. (31) found that SR, as measured by SIS, was significantly better at 12 months in comparison with 3 months post-stroke. This could not be

explained by level of dependency, age or sex, nor could it be explained by changes in other SIS domains. In the current study the outcome of SR could not be explained by age, sex, type of stroke, CVD, or living arrangements. A significant correlation was, however, found between perception of overall SR and higher level of PA. This once again highlights the importance of ensuring stroke survivors achieve an adequate level of PA even long-term post-stroke.

All 3 areas of SIS that the present study examined demonstrated a positive relationship between a higher level of PA and a lower level of perceived impact of stroke on life. As mentioned in the background, the stroke population have been found to have a higher energy expenditure compared with the normal population (18). However, our findings oppose the hypothesis that higher levels of PA would lead to greater energy loss, increased fatigue, and thus increased impact of stroke on life.

Multidisciplinary rehabilitation post-stroke has also been shown to be essential in order to minimize long-term symptoms and is optimized with early team input (32). Furthermore, in order to improve long-term levels of PA post-stroke, individualized and clientcentred interventions have been found to be beneficial (6). Stroke rehabilitation is complex, and this study's findings support a continued emphasis on assisting stroke survivors to achieve an adequate level of PA long-term post-stroke.

This study has some limitations. It is important to recognize that variables other than those examined may have impacted on outcomes, such as cultural and religious differences. The SALGOT study was an observational study with participants with initially impaired upper extremity function, who received standard Swedish rehabilitation and care. This may have influenced results and should be kept in mind when interpreting the results of this study. A larger study sample could have made it possible to complete analysis of all intended SIS domains and could have minimized any significant differences between included and excluded groups. The participants included in the study were younger than the non-included, which could be explained by the study being conducted 12 months post-stroke and older people generally having a lower survival rate post-stroke (33). However, elderly people are also found to be less active in general (34), and thus with a comparatively younger study group this could affect the results.

Conclusion

The results of this study show that PA at 12 months post-stroke has a significant and positive correlation

with perceived strength and stroke recovery. Furthermore, PA combined with level of dependency is significantly and positively correlated with perceived participation at 12 months post-stroke. This is critical information within rehabilitation, and particularly within physiotherapy, where much focus is placed on encouraging and assisting stroke survivors to increase their level of PA. This is an important clinical finding, as it provides evidence for encouraging an active lifestyle post-stroke, with the knowledge that it can help survivors to increase their perceived strength, participation and stroke recovery post-stroke.

ACKNOWLEDGEMENTS

This project was funded by Greta and Einar Askers Foundation and The Local Research and Development Board for Gothenburg and Södra Bohuslän. The SALGOT study was supported by grants from the Swedish Research Council (VR 2012-70X-22122-01-3), the Swedish Research Council (VR 2017-00946), the Swedish Heart-Lung Foundation, the Swedish Brain Foundation, Promobilia, and the Swedish Stroke Association.

The authors would like to thank Margit Alt Murphy and Eva-Lena Bustrén for their work with data collection and the Riks-Stroke Collaboration for help with demographic data.

The authors have no conflicts of interest to declare.

REFERENCES

- World Health Organization. Media centre. Fact sheet: physical activity. 2018 Feb 23 [cited 2019 Sept 16]. Available from: http://www.who.int/mediacentre/factsheets/ fs385/en/.
- Singh R, Pattisapu A, Emery MS. US Physical activity guidelines: current state, impact and future directions. Trends Cardiovasc Med 2019 Oct 17 pii: S1050-1738(19)30140-9 [Epub ahead of print].
- English C, Manns PJ, Tucak C, Bernhardt J. Physical activity and sedentary behaviors in people with stroke living in the community: a systematic review. Phys Ther 2014; 94: 185–196.
- World Health Organization. Health topics: stroke, cerebrovascular accident. 2017 [cited 2019 Sept 16]. Available from: http://www.who.int/topics/cerebrovascular_accident/ en/.
- English C, Hillier SL, Lynch EA. Circuit class therapy for improving mobility after stroke. Cochrane Database Syst Rev 2017; 6: CD007513.
- Morris JH, Macgillivray S, McFarlane S. Interventions to promote long-term participation in physical activity after stroke: a systematic review of the literature. Arch Phys Med Rehabil 2014; 95: 956–967.
- Flansbjer UB, Miller M, Downham D, Lexell J. Progressive resistance training after stroke: effects on muscle strength, muscle tone, gait performance and perceived participation. J Rehabil Med 2008; 40: 42–48.
- Saunders DH, Sanderson M, Hayes S, Kilrane M, Greig CA, Brazzelli M, et al. Physical fitness training for stroke patients. Cochrane Database Syst Rev 2016; 3: CD003316.
- Danielsson A, Meirelles C, Willen C, Sunnerhagen KS. Physical activity in community-dwelling stroke survivors and a healthy population is not explained by motor function only. PM R 2014; 6: 139–145.
- 10. Danielsson A, Willen C, Sunnerhagen KS. Physical acti-

p. 6 of 6 P. Cook et al.

vity, ambulation, and motor impairment late after stroke. Stroke Res Treat 2012; 2012: 818513.

- Mudge S, Barber PA, Stott NS. Circuit-based rehabilitation improves gait endurance but not usual walking activity in chronic stroke: a randomized controlled trial. Arch Phys Med Rehabil 2009; 90: 1989–1996.
- World Health Organization. International Classification of Functioning, Disability and Health: ICF. Geneva: WHO Library Cataloguing-in-Publication Data; 2001.
- Goljar N, Burger H, Vidmar G, Marincek C, Krizaj J, Chatterji S, et al. Functioning and disability in stroke. Disabil Rehabil 2010; 32: S50–S58.
- Duncan PW, Bode RK, Min Lai S, Perera S. Rasch analysis of a new stroke-specific outcome scale: the Stroke Impact Scale. Arch Phys Med Rehabil 2003; 84: 950–963.
- Mossey JM, Shapiro E. Self-rated health: a predictor of mortality among the elderly. Am J Public Health 1982; 72: 800–808.
- Lai SM, Studenski S, Richards L, Perera S, Reker D, Rigler S, et al. Therapeutic exercise and depressive symptoms after stroke. J Am Geriatr Soc 2006; 54: 240–247.
- Studenski S, Duncan PW, Perera S, Reker D, Lai SM, Richards L. Daily functioning and quality of life in a randomized controlled trial of therapeutic exercise for subacute stroke survivors. Stroke 2005; 36: 1764–1770.
- Platts MM, Rafferty D, Paul L. Metabolic cost of over ground gait in younger stroke patients and healthy controls. Med Sci Sports Exerc 2006; 38: 1041–1046.
- Alt Murphy M, Persson HC, Danielsson A, Broeren J, Lundgren-Nilsson A, Sunnerhagen KS. SALGOT – Stroke Arm Longitudinal study at the University of Gothenburg, prospective cohort study protocol. BMC Neurol 2011; 11: 56.
- Grimby G. Physical activity and muscle training in the elderly. Acta Med Scand Suppl 1986; 711: 233–237.
- Grimby G, Frandin K. On the use of a six-level scale for physical activity. Scand J Med Sci Sports 2018; 28: 819–825.
- Asplund K, Hulter Asberg K, Appelros P, Bjarne D, Eriksson M, Johansson A, et al. The Riks-Stroke story: building a sustainable national register for quality assessment of stroke care. Int J Stroke 2011; 6: 99–108.

- Perna R, Temple J. Rehabilitation outcomes: ischemic versus hemorrhagic strokes. Behav Neurol 2015; 2015: 891651.
- 24. Kwakkel G, Kollen BJ. Predicting activities after stroke: what is clinically relevant? Int J Stroke 2013; 8: 25–32.
- 25. Quinn JT, Dawson RJ, Walters RM, Lees RK. Reliability of the Modified Rankin Scale: a systematic review. Stroke 2009; 40: 3393–3935.
- Reeves MJ, Bushnell CD, Howard G, Gargano JW, Duncan PW, Lynch G, et al. Sex differences in stroke: epidemiology, clinical presentation, medical care, and outcomes. Lancet Neurol 2008; 7: 915–926.
- Fernandez D, Brotons C, Moral I, Bulc M, Afonso M, Akan H, et al. Lifestyle behaviours in patients with established cardiovascular diseases: a European observational study. BMC Fam Pract 2019; 20: 162.
- 28. Van Domelen DR, Koster A, Caserotti P, Brychta RJ, Chen KY, McClain JJ, et al. Employment and physical activity in the U.S. Am J Prev Med 2011; 41: 136–145.
- Tornbom K, Persson HC, Lundalv J, Sunnerhagen KS. Self-assessed physical, cognitive, and emotional impact of stroke at 1 month: the importance of stroke severity and participation. J Stroke Cerebrovasc Dis 2017; 26: 57–63.
- Pallant J. SPSS survival manual : a step by step guide to data analysis using IBM SPSS. 5th edn. Maidenhead: Maidenhead : McGraw-Hill; 2013.
- Guidetti S, Ytterberg C, Ekstam L, Johansson U, Eriksson G. Changes in the impact of stroke between 3 and 12 months post-stroke, assessed with the Stroke Impact Scale. J Rehabil Med 2014; 46: 963–968.
- Stroke Unit Trialists' Collaboration. Organised inpatient (stroke unit) care for stroke. Cochrane Database Syst Rev 2013; (9): CD000197.
- Riksstroke. [Stroke and TIA: Riksstroke annual report 2015.] Umeå: Riksstroke, 2016 [cited 2017 Nov 10] (in Swedish).
- Roberts CE, Phillips LH, Cooper CL, Gray S, Allan JL. Effect of different types of physical activity on activities of daily living in older adults: systematic review and meta-analysis. J Aging Phys Act 2017; 25: 653–670.